Author’s response to reviews

Title: Can opportunities be enhanced for vaccinating children in home visiting programs? A population-based cohort study.

Authors:

Michael R Isaac (Michael.Isaac@gov.mb.ca)
Allan Katz (Akatz@cpe.umanitoba.ca)
Carole Taylor (Carole_Taylor@cpe.umanitoba.ca)
Chun Yan Goh (ChunYan_Goh@cpe.umanitoba.ca)
Dan Chateau (Dan_Chateau@cpe.umanitoba.ca)
Elaine Burland (Elaine_Burland@cpe.umanitoba.ca)
Joykrishna Sarkar (Joykrishna_Sarkar@cpe.umanitoba.ca)
Mariette Chartier (Mariette_Chartier@cpe.umanitoba.ca)
Marni Brownell (Marni_Brownell@cpe.umanitoba.ca)
Nathan C Nickel (Nathan_Nickel@cpe.umanitoba.ca)
Patricia J Martens (Pat_Martens@cpe.umanitoba.ca)
Milton Hu (Milton_Hu@cpe.umanitoba.ca)

Version: 2

Date: 21 January 2015

Author’s response to reviews:

Included new information in the Methods section:

"Data for this study were derived from the PATHS Data Resource, a unique database developed for the PATHS program of research containing anonymized individual-level data from the Population Health Research Data Repository, which is housed at the Manitoba Centre for Health Policy (MCHP). [19] The PATHS Data Resource includes data provided by Manitoba Health, Healthy Living and Seniors; Manitoba Jobs and the Economy; Manitoba Family Services; Healthy Child Manitoba; the Winnipeg Regional Health Authority; Statistics Canada; and Manitoba Housing and Community Development. The data were not 'openly available', but rather permission for its use was requested and received from each of the individual data providers, as well as the Health Information Privacy Commission (HIPC) of the Government of Manitoba. However, other researchers interested in using these data are encouraged to apply for access; more information on applying for access can be found elsewhere. [19]"